Author's response to reviews

Title: Plasmodium falciparum parasitaemia and malaria in pregnant women at first clinic visit in the mount Cameroon Area

Authors:

   Judith K Anchang-Kimbi (kuoh2000@yahoo.fr)
   Vera N Nkweti: (vepretty89@yahoo.com)
   Helen N Ntonifor (ngumnto@yahoo.com)
   Tobias O Apinjob (apinjohtoby@yahoo.co.uk)
   Rolland B Tata (rollandbantar@yahoo.com)
   Hanesh C Fru (hanesh_ch@yahoo.com)
   Eric A Achidi (achidi_e@yahoo.com)

Version:7 Date:1 September 2015

Author's response to reviews: see over
Dear Editor,

We are re-submitting for consideration for publication our revised manuscript entitled: ‘Plasmodium falciparum parasitaemia and malaria among pregnant women at first clinic visit in the mount Cameroon Area’ by Anchang-Kimbi JK, Nkweti VN, Ntonifor HN, Apinjoh TO, Tata RB, Chi HF and Achidi EA. We thank the BMC Infectious Diseases Journal for giving us the opportunity to revise our manuscript and it is apparent to us that the revisions have significantly improved the quality of the manuscript.

We have made the corrections and replied to the comments suggested by the editor in the forwarded ms text attached to this mail. As requested, we have responded point by point to the editor’s comments as indicated below. I wish to inform the Editor that all the authors have read and approved the revised version of the manuscript and authorized me to submit it on their behalf.

We look forward to a fruitful approval of the manuscript.

Thank you.

Yours sincerely
Anchang-Kimbi JK
Response to Editor’s Comment

The limitations of the study are clearly outlined in the discussion section (lines 363 – 379)